Author's response to reviews

Title: Influences of socioeconomic factors on childhood and adolescent overweight by gender in Korea: cross-sectional analysis of nationally representative sample

Authors:

Jin-Won Noh (jinwon.noh@gmail.com)
Young-eun Kim (kimyes@neca.re.kr)
In-Hwan Oh (parenchyme@gmail.com)
Young Dae Kwon (snukydl1@naver.com)

Version: 5
Date: 20 March 2014

Author's response to reviews: see over
Dear Associate Editor,

Thank you for the kind and thoughtful opinion. It makes our manuscript more interesting and meaningful. We tried to revise the manuscript upon your advices. Here below is the explanation giving a point-by-point response to the concerns. The changes to our manuscript are marked using blue color.

Subject: MS: 5422170161117657 - Influences of socioeconomic factors on childhood and adolescent obesity by sex in Korea: cross-sectional analysis of nationally representative sample

Minor revisions:

- Please change description of the study to past tense for consistency:
  -> We changed the description to past tense.

- Please use "data" as a plural.
  -> We used data as a plural in the manuscript.

Sincerely,

Young Dae Kwon, MD, PhD